Jour Fixe

SS, WFP, PG

Atomslab

• AC was fixed by the 4440 guys \rightarrow filter has to be changed once a year

Quench

- The alignment of the fiber from the laser further dropped to 570 mW out of 1400 mW.
- Alignment of AOM in progress

PTB/Meta-mirror

- Reflectivity of Meta-mirror was measured to be 99.8x% via Ring-Down
- Measurement of transmission is planned (preperation in progress)
- (time-consuming) Tender-process for Wavemeter

From: https://iqwiki.iqo.uni-hannover.de/ - **IQwiki**

Permanent link: https://iqwiki.iqo.uni-hannover.de/doku.php?id=groups:mg:private:meetingnotes:2020-09:meetingnotes-2020-09-24

Last update: 2020/09/29 11:40

